Author's response to reviews

Title: Improving healthcare consumer effectiveness: An Animated, Self-serve, Web-based Research Tool (ANSWER) for people with early rheumatoid arthritis

Authors:

Linda C Li (lli@arthritisresearch.ca)
Paul Adam (Paul.Adam@vch.ca)
Anne F Townsend (atownsen@interchange.ubc.ca)
Dawn Stacey (Dawn.Stacey@uOttawa.ca)
Diane Lacaille (dlacaille@arthritisresearch.ca)
Susan Cox (suecox@interchange.ubc.ca)
Jessie McGowan (jmcgowan@uottawa.ca)
Peter Tugwell (ptugwell@uottawa.ca)
Gerri Sinclair (gerris@gnwc.ca)
Kendall Ho (kendall.ho@ubc.ca)
Catherine L Backman (catherine.backman@ubc.ca)

Version: 3 Date: 12 August 2009

Author's response to reviews:

August 11, 2009

Dear BioMed Central Editorial Production Team,

RE: Improving healthcare consumer effectiveness: An Animated, Self-serve, Web-based Research Tool (ANSWER) for people with early rheumatoid arthritis

Enclosed please find the revision of the abovementioned paper. The following changes have been made:

1. The abstract is now structured into Background, Methods/Design, Discussion.
2. The former Reference 24 is now included in the text.
3. The former Reference 22, 23, 27 have been deleted. They are replaced by the funding agency’s web addresses where summaries of the research proposal can be found.
4. The page numbers have been added to the former Reference 41 (now Reference XX). Reference 4 and 5 are full Cochrane reviews and do not have volume or page number.
5. Author details and place of publication are now provided for the former Reference 42 (now Reference XX)
6. All authors’ names are listed in references with 10 or fewer authors.
7. All underlining have been changed to bold.
8. Additional files that contain ethical approval and cover letter have been
deleted.

Sincerely yours,
Linda Li, PT, PhD